Author's response to reviews

Title: DALI: Defining Antibiotic Levels in Intensive care unit patients: A multi-centre point of prevalence study to determine whether contemporary antibiotic dosing for critically ill patients is therapeutic

Authors:

Jason A Roberts (j.roberts2@uq.edu.au)
Jan J DeWaele (Jan.DeWaele@UGent.be)
George Dimopoulos (gdimop@med.uoa.gr)
Despoina Koulenti (deskogr@yahoo.gr)
Claude Martin (claude.martin@mail.ap-hm.fr)
Philippe Montravers (philippe.montravers@bch.ap-hop-paris.fr)
Jordi Rello (jrello@crips.es)
Andrew Rhodes (andrewrhodes@nhs.net)
Therese Starr (Therese_Starr@health.qld.gov.au)
Steven C Wallis (s.wallis@uq.edu.au)
Jeffrey Lipman (j.lipman@uq.edu.au)

Version: 2 Date: 18 June 2012

Author's response to reviews:

Dear Sir/Madam,

Please find included with this resubmission, the document that describes the country and city names of each of the participating hospitals in the DALI Study as well as the Ethical Committee that has approved conduct of this study.

In line with the email request for the above information from the journal, no changes have been made to the manuscript itself.

Please do not hesitate to contact me if any further information relating to this study is required.

Yours sincerely

Jason Roberts